family goods and services. Washington, DC: American Home Economics Association, 1976.

WEISS, YORAM. "The Determination of Life Cycle Earnings: A Survey," in Handbook of labor economics, Vol. I. Eds.: ORLEY C. ASHENFELTER AND R. LAYARD. Amsterdam: North-Holland, 1986, pp. 603-40.

WILLIS, ROBERT. "A New Approach to the Economic

Theory of Fertility Behavior," J. Polit. Econ., Mar./Apr. (2, part 2) 1973, 81, pp. S15-64.

- YELLEN, JOHN E. "The Transformation of the Kalahari !Kung," Scientific American, Apr. 1990, 262(4), pp. 96–105.
- ZICK, CATHLEEN AND BRYANT, W. KEITH. "Alternative Strategies for Pricing Home Work Time," *Home Econ. Res. J.*, 1983, 12(2), pp. 133-44.

MAMIO' EJPERD QUE TE INTEREJE TE TNTEREJE

Journal of Economic Literature Vol. XXIX (June 1991), pp. 523–552

The Health Care Quadrilemma: An Essay on Technological Change, Insurance, Quality of Care, and Cost Containment

By BURTON A. WEISBROD Northwestern University

I want to thank a number of people. Paul Boben, Ted Ingham, and Elizabeth Mauser provided excellent research assistance. Christoph Badelt, David Cleeton, Bruno Frey, W. Lee Hansen, Robert Haveman, Bryan R. Luce, Todd Sandler, Mark Satterthwaite, Barbara Wolfe, participants in my seminar at the Rand Corporation, and three anonymous referees gave valuable comments. Judith Kirkwood was an outstanding editor.

I. Introduction

URING THE ROUGHLY FOUR DECADES since the end of World War II, the health care system in the United States has experienced historically unprecedented change in three dimensions. First, new technologies have revolutionized the ways in which health care is capable of being practiced. Almost all of today's armamentarium of disease diagnosis and treatment devices and techniques were unknown 40 years ago. In the case of prescription drugs, for example, about 10 percent of the 200 largest-selling drugs are new each year; and only 25 percent of the 200 top-selling drugs in 1972 remained in the group 15 years later (David Cleeton, Valy Goepfrich, and Burton Weisbrod 1990).

Second, the role of health care insurance—private and public—has expanded dramatically. By 1980, 82.5 percent of the U.S. population had some health care insurance, compared with fewer than 10 percent in 1940.¹

Third, personal health expenditures have soared. From \$300 per capita in 1950, they leaped to \$1,493 in 1987 (all in 1982 dollars). The percentage of GNP

¹Throughout the postwar period the expansion of private health care insurance has been spurred by federal tax policy. By making employer-financed health insurance nontaxable income to employees, federal policy distorted worker choice between health insurance and cash wages, encouraging excess health insurance (Martin Feldstein and Elizabeth Allison 1974; Mark Pauly 1974; Bridger Mitchell and Ronald Vogel 1975; Mitchell and Charles Phelps 1976; Amy Taylor and Gail Wilensky 1983; Howard Chernick, Martin Holmer, and Daniel Weinberg 1987). devoted to medical care has almost tripled over that period—from 4 to 11 percent (U.S. Bureau of the Census 1979, p. 97; U.S. Bureau of the Census 1989, p. 90; Suzanne Letsch, Katherine Levit, and Daniel Waldo 1988).

This paper explains how the expansion of health care insurance has paid for the development of cost-increasing technologies, and how the new technologies have expanded demand for insurance. My goal is less to review the vast literature on the health care system and the rising level of real expenditures on it, than to reflect on the dynamic interplay of incentives for the R & D sector to develop particular kinds of new technologies, the role of the insurance system in that process, and, reciprocally, the long-run effects of new technologies (any new knowledge about health care) on the character of the health care insurance system. The broad model outlined here highlights the ways in which the quality of health care that is technically feasible to supply at any point in time, and the breadth of access to that care, influence each other and the aggregate level of health care expenditures, but the model is not fully specified, nor is it tested rigorously. Thus, this essay should be seen as a personal interpretation-largely positive, rather than normative, in character-of a period of enormous growth and massive change in both the practice and finance of health care.

The central focus on technological change—as an independent variable causing changes in the form and extent of insurance coverage, and as a dependent variable, being influenced by incentives operating through the health insurance system—highlights the impact of incentives; both the pace and types of research and development are functions of rewards that are endogenously variable, as are the comprehensiveness of insurance coverage and the breadth of ac-

. .

cess to it.² The following propositions are set forward: (1) The amount of resources going into the R & D process, and its direction, during some time interval._depend in part on the mechanisms expected to be used to finance the provision of health care in future periods, when the fruits of the research process become marketable. This is simply to say that R & D is influenced by expected utilization, which depends on the insurance system. Reciprocally, (2) the demand for health care insurance depends, in part, on the state of technology, which reflects R & D in prior periods. These relationships help to explain why (3) long-run growth of health care expenditures is a by-product of the interaction of the R & D process with the health care insurance system.³ I also examine briefly some effects of alternative forms of health care insurance on the quality of care, as distinguished from its quantity, and long-run changes in the definition of "health care" under insurance, as endogenous R & D alters the menu of technically feasible measures.

To understand the markets in which health care is provided and financed, it is useful to consider ways in which health care differs from most other commodities. First, it sometimes involves the preservation of life, or, at least, major effects on the quality of life. Second, it is a technically complex commodity that abounds with informational asymmetries, adverse to consumers (Kenneth Arrow 1963; George Akerlof 1970; Richard Titmuss 1971). Third, and as a result of

² Other effects of health insurance, particularly on incentives for utilization of health services, have received considerable attention. For a recent and valuable review see Pauly (1986).

³ Other forces also affect health care expenditures. Rising real income appears to have a positive effect on demand for health care; an income elasticity of ± 0.2 (or less) has been estimated from the Rand health insurance experiment (Willard Manning, et al. 1987). these two characteristics, "nonmarket" (governmental and private nonprofit) suppliers in the health care sector, especially among hospitals, nursing homes, and blood banks, play a large role in influencing the interaction between insurance and R & D.⁴

Because health care affects length and quality of life, many societies have come to accept the normative proposition that "high"-quality care ought to be made available widely, regardless of an individual's ability to pay. This assignment of property right-the breadth of which is under continuing debate-results in pressure on government to finance access to some health care redistributively. In the U.S., private market financing of health care, by individuals and employers, has been supplemented by governmental resources-particularly through the Medicare and Medicaid programsand to a smaller extent, through private charitable activities.

Another reason---in addition to provid-ing widespread access-for society's willingness to intervene in private health care markets is the substantial informational asymmetries, which give rise to economic and political demands for consumer protection (Arrow 1963; Weisbrod 1978, 1989; Henry Hansmann 1980). The claims that physicians "induce" demand (Arrow 1963; Robert Evans 1974; Gail Wilensky and Louis Rossiter 1983; Rossiter and Wilensky 1984; Uwe Reinhardt 1985; Jerry Cromwell and Janet Mitchell 1986; Miron Stano 1987), that they engage in "defensive medicine"-diagnostic testing and other practices that have no expected benefits for patient health but are defenses in "malpractice" suits

⁴ Some readers may prefer the term *nonprofit* to *nonmarket*. Whatever term is used, the point is to distinguish private, profit-oriented organizations from the institutions of either government or the private nonprofit sectors. To be sure, government and private nonprofit organizations operate in "markets," in the sense that exchange occurs.

(Mohan Garg, Werner Gliebe, and Mounir Elkhatib 1978; Stephen Zuckerman 1984; Patricia Danzon 1985)-and that they perform "unnecessary" surgery⁵ may or may not be valid; they are plausible, however, only if physicians are better informed than their patients (Pauly 1979) and do not act as perfect agents.⁶ The importance of health care to life and well-being, combined with the limited ability of consumers to make well-informed judgments about quality of care, and with imperfect agency relationships with physicians may help to explain why consumers of health care rely upon public and private nonprofit institutions to an unusual degree.

The remainder of the paper proceeds as follows: Part II contains a brief outline of the recent history of the health care sector in the United States-its evolving technology, changing insurance/finance system, increasing level of real health care expenditures; and the advent of cost control measures. Part III shows how the constellation of services included in "health care" is endogenous, being affected by the interaction of the insurance system and the R & D process. Part IV focuses on the effects of R & D (technological change) on the health care insurance system. Part V looks at the reciprocal effects of the insurance system on the R & D sector. Part VI deals with the effects of alternative insurance systems on quality of care, with the state of technology fixed, and Part VII summarizes and points up some possible generalizations beyond health care.

Finally, examining these interdepen-

⁶ Operationalizing the concepts of "induced" demand, "defensive" medicine, and "unnecessary" surgery—each of which reflects a market failure to the extent it occurs—poses serious problems. These issues, however, are beyond the scope of this paper.

⁵ A congressional subcommittee estimated that in 1977 there were 2 million unnecessary operations, at a cost of \$4 billion and with a loss of 10,000 lives ("Elective Surgery: Cut it Out" 1979).

dent relationships may help to explain some of the differences across countries in financing of health care and their roles in health care R & D, for the forces at work are not uniquely North American and the policy implications can be generalized. The U.S. is unusual, however, in the extent to which its actions as a producing and a consuming country influence the rate and direction of health care R & D. No other country is so major an actor in both the R & D (producing) sector and the health care (consuming) sector. For most other countries, outputs of the R & D sector are essentially exogenous to their methods of financing health care, and their systems of health care finance are also essentially exogenous to their own R & D activities. Switzerland, for instance, is a substantial producer of health care R & (especially pharmaceuticals), but "+ , a small consumer; th Unitea . .om and Japan, althous they are not the 'al elements in the R & D sector, are ----Jut puts of that sector. ... ormou size and therefore impact of both the producing and consuming elements in the United States that make it such a fine subject for study.

II. *if cent History of Health* Care *ted States: Technological* Change and the Growth of Insurance Coverage

One striking aspect of change in the U.S. health care system since World War II has been the dramatic increase in knowledge of means for diagnosing and treating illness. Fifty years ago, physicians were little more than diagnosticians, their activities being essentially "limited to identification of . . . illness, the prediction of the likely outcome, and

then the guidance of the patient and his family while the illness ran its full, natural course" (*Report of the President's Biomedical Research Panel* 1976, appendix A, p. 3). Today, the scope of effective interventions includes kidney dialysis, organ transplants, polio vaccines, arthroscopic surgical techniques, CT scanners, nuclear magnetic resonators, in vitro fertilization. As recently as a decade ago, heart and liver transplants were virtually unknown, but their numbers have soared, from 62 and 26 in 1981 to 1,441 and 1,182 in 1987, respectively (U.S. Bureau of the Census 1989, table 166).

At the same time that the technology of health care has been changing so dramatically, t^1 system for financing health care has been revolutionized. In the quarter century be ween 1950 and 1973 alone, the share of health care expendi-

.s that was met by insurance more than tripled, from 12 to 41 percent (U.S. Bureau of the Census 1975, table 105). The mix of private and governmental insurance also changed during that period; while total private expenditures on health and medical services were growing almost sixfold, from \$8.7 billion to \$59.8 billion (current dollars), government expenditures (Medicare and, to some extent, Medicaid) were leaping fourteenfold, from \$2.5 billion to over \$37 billion (U.S. Bureau of the Census 1975, table 100). Insurance coverage for "major" or "catastrophic" health care costs has also risen sharply, from 22 percent of the population in 1960 to 73 percent by 1984 (U.S. Bureau of the Census 1987, tables 1, 2, and 137).

Initially, most health insurance was of one particular type, covering a limited menu of only hospital services—perhaps after a small deductible—and paying ("reimbursing") the hospital for the particular services provided to a patient, the payment being equal to the "actual" average cost of treating that patient with

whatever technology was used (Rosemary Stevens 1989). Included was an approximation of the average variable cost of any diagnostic or therapeutic procedures performed on the patient's behalf, plus a per diem payment for room, board, and basic nursing services, and, in the case of for-profit hospitals, a markup. Thus, the payment received by the hospital was determined retrospectively and was a function of endogenous decisions by the hospital and physician as to length of stay and the resources deployed in treating each specific patient. With hospital revenue being a function of the cost of services provided, there was little incentive to weigh costs against patient benefits. Any diagnostic or therapeutic resource that had a positive expected value of benefits was financially feasible to provide, and even when there was great uncertainty about the probability distribution of benefits from a new, more costly technology, the absence of a budget constraint encouraged its adoption.

By the 1970s, however, the growth of real expenditures on medical care-reflected in rising private insurance premiums, Medicare budgets, and the share of GNP devoted to health care-had become matters of growing public concern. Some attributed this "health care cost inflation" to the insurance system and its effect on demand; retrospective payment arrangements, operating through the insurance system, were encouraging "overuse" of medical resources (M. Feldstein and Bernard Friedman 1977; Pauly 1986). The result was a spate of reforms designed to force health care providers to consider the cost consequences of their decisions. This was done by making more of providers' revenue "prospective." HMOs (Health Maintenance Organizations) and, beginning in October 1983, the Medicare DRG (Diagnosis-Related Groups) system for pricing hospital services, are the preeminent examples of this type of reform.

Both HMOs and the Medicare prospective-payment system confront suppliers with the incentive to be more costconscious, but they differ in the comprehensiveness of that incentive. Under the current DRG system for paying hospitals, the "fixed" payment for a particular patient is supplemented by additional payments to cover capital costs; thus, there is some incentive for hospitals to substitute capital for labor.⁸ In addition, under the DRG system, as under the previous retrospective-pricing system, a hospital's revenue is a function of its admissions of patients; this produces an incentive to hospitalize rather than to utilize approaches that involve nonhospital inputs such as drugs, broad medical management approaches, and instruction of patients in ways to prevent and alleviate problems through life-style and dietary measures. HMOs, which have a contractual responsibility to provide medical services, not simply hospital treatment, and receive a flat annual fee per member, maintain a greater financial incentive to utilize alternatives to hospitalization.

To the extent that cost-based insurance has been at the root of the rising expenditures on health care, however, the causal mechanism is less clear than it seems. The moral hazard effect of insurance could cause patients and their physicianagents to utilize more health care resources, and therefore aggregate health care expenditures to be greater than they would otherwise be; yet it does not follow that insurance would cause expenditures on health care to grow more rapidly. Something had to be changing. That "something" could have been the state of technology which, as we will see, was expanding in a systematic direction as a consequence, at least in part, of the par-

⁸ I owe this point to an anonymous referee,

For a broader, European, perspective on health care systems, see Organization for Economic Co-operation and Development (1990).

ticular form of insurance that had been adopted. An expanding health care insurance system—more widespread coverage of people and broader coverage of health care resources such as pharmaceuticals and chiropractic services—might also account for growth of health care expenditures, but this explanation would pose the question of why insurance coverage would be expanding.⁹

The major theme of this paper is that the demand for health care insurance and the process of technological change are interdependent. A shift away from insurance that paid hospitals and physicians on the basis of endogenously determined "costs incurred" and office visits, to insurance that paid amounts that were largely independent of costs incurred on behalf of any particular patient, represented a major change. It altered incentives to use existing health care resources (that is, their rate of diffusion and utilization) and it altered incentives for the R & D sector to invest in developing medical care techniques that were of higher quality but more costly.

As noted above, the shift in the nature of health insurance has occurred in two principal forms-expansion of HMOs and adoption of the DRG system of hospital pricing. In the decade of the 1980s alone, enrollments in HMOs more than tripled, from 9.1 million in 1980 to 28.6 million in 1987 (U.S. Bureau of the Census 1989, table 148). Under the DRG prospective-payment system, a hospital receives payment (prices) for treatment (e.g., of appendicitis) based on industrywide costs for each of the 468 DRG categories. Thus, conditional on admission of a patient with a particular diagnosis, what a hospital faces is a price for treatment that is essentially independent of the actual resource cost it incurs (C. Hogan 1988).¹⁰

Both HMOs and the DRG system of pricing hospital services are potentially revolutionary in their incentive effects on R & D.11 The fact that the principal objective of each of these forms of prospective pricing was fiscal control is not in doubt (Pauly 1986). Several related matters, however, are far from clear and deserve more research: Why did the shift in insurance mechanisms, from retrospective to prospective, occur when it did? Why did the United States ever start with insurance based on retrospective and fee-for-service pricing; after all, the incentives that cost-based pricing generated were, or at least should have been, apparent long ago, and the fiscal problem, as manifested in the rising share of GNP devoted to health care, has been growing for decades.

In some current research, Paul Boben (1989) presents a model in which retrospective pricing of hospital services and physician services (through fee-for-service payments to physicians on the basis of "usual and customary" fees) is allocatively efficient when there is little insurance coverage and health care prices are determined in relatively competitive markets, but diminishes as that coverage spreads. In this model the discipline of prices on patient and provider behavior that prevails when few people have insurance gives way to growing price insensitivity (inelasticity) with the expansion of insurance. Thus, a "tipping-point" is

¹⁰ The pricing system is not entirely rigid. For example, a hospital may collect from Medicare more than the DRG price for a limited number of unusually high-cost "outliers."

¹¹ The DRG system of hospital service pricing initially applied only to Medicare patients. It has subsequently been expanded, however, through private arrangements, to a growing number of other patients who are not covered by the Social Security Medicare law: reached, at which the usefulness of market-determined prices as signals of opportunity costs becomes less than its cost in terms of distorted resource allocations (the moral hazard problem). Such modeling of the social choice of the insurance system is in its infancy.

Many of the issues raised above have received scant attention in the literature. The effect of advancing technology on health care financing arrangements, the incentives for research and development inherent in those financial arrangements, and the implications of those arrangements for the quality of the care provided, are each the subject of later sections, where we will also consider the inevitability that health care expenditures would soar in the post-World War II era. But, first, how do we define "health care"? How is it affected by technological change and how does its definition affect insurance coverage?

III. Defining "Health Care"

Up to this point we have been discussing the market for "health care" without defining that market carefully. The endogeneity of the definition of health care under insurance contracts has received some attention (John Goddeeris 1984a, 1984b). Consider two nonmutually exclusive hypotheses concerning the causes and consequences of the definition of health care under insurance: (1) The operational definition of health care, under insurance contracts, is a function of the state of medical technology; (2) the state of medical technology today is a function of economic and political responses to prior definitions of health care coverage under insurance.

The way health care is defined under insurance contracts is important for a number of reasons, positive and normative. It affects the level of insured expenditures, the incentives to utilize resources that are covered relative to those that are not (Paul J. Feldstein 1988), and the incentives for the R & D sector to explore various potential health-promoting technologies. At the operational level, the definition of health care is at issue when coverage for chiropractic care or for "experimental" drugs or other "new" technologies is debated.

The effect of health care insurance on incentives for R & D depends on the operational definition of health carethat is, on the boundaries of the insurance contract. Health insurance contracts do not offer the option of coverage only for particular subsets of technologies, such as those already available at a given point in time (Goddeeris 1984b; Goddeeris and Weisbrod 1985; James Baumgardner 1989). A reasonable conjecture, however, is that health care expenditures today would be substantially lower than they are if health care were being defined, for insurance purposes, as limited to the use of medical technologies available at the time the policy took effect, or at some other fixed date. The more broadly health care is interpreted under the contract, and the more responsive it is to changes in technology, the broader the range of activities over which insurance will encourage R & D.

What determines how health care is defined? I suggest that the R & D process causes the definition of what is covered by health insurance to change in systematic ways. Technological advances are not only expanding the range of medical capabilities for extending life and enhancing health status, as the latter term is customarily understood; they are also presenting opportunities to deal with problems not conventionally considered to be "illnesses," in ways not conventionally considered "health care."¹²

¹² Another example of the need to decide, as a matter of public policy, how to define operationally what is health care involves people with physical dis-

⁹ Even with constant technology, real costs of health care could increase if input prices rose—for example, because of increased unionization of hospital labor, and this could increase the demand for insurance, ceteris paribus.

An illustration of this causal process is the current debate over whether health insurance should necessarily cover in vitro fertilization. This has become an issue only in the past few years, when advances in medical capabilities made such fertilization technically feasible. An advance in medical technology has led to pressure to expand the traditional definition of insurance coverage, pressure being felt now through the political system; by 1988 such insurance coverage had been mandated in five states (U.S. Congress 1988),¹³ and by the end of 1989, laws requiring insurers to cover such "advanced" treatments for infertility had been er stad in " nd bills had been ers (Sonia Nazario 1989).

The effect of technological change on the health insurance market can also be seen with "experimental" drugs. The decision to term a drug experimental is often seen as a statement of the degree of professional knowledge about its safety and efficacy. It is, however, also a statethe hether the drug will or will ment "health care" for insurnot l use insurance typiance purpose cally does not cove. perimental" technologies. For example as long as the AIDS drug, AZT, was termed experimental, its exclusion from coverage under health insurance involved each patient with costs that, until 1990, have been in excess of \$8,000 per year, even

. .

٤,

abilities. Surgery and physical therapy illustrate "traditional" health care resources employed to reduce the disabilities. City buses that are wheelchair-accessible are unquestionably valuable to the disabled; whether their cost should be regarded as health care expenditures and covered by health care insurance is another matter.

¹³ As of May 1988, Arkansas, Hawaii, Maryland, Massachusetts, and Texas had enacted legislation requiring private insurers to provide some coverage for in vitro fertilization procedures. Delaware Blue Cross/Blue Shield began offering coverage voluntarily in response to legislative activity (U.S. Congress 1988) though conventional hospital-based treatment was covered in traditional fashion.

The hypothesis that the definition of health care is endogenous to the economic-political system in which health care insurance is defined, provided, and financed has important implications, to the extent it is valid. If insurance coverage is defined, as it has been, to encompass new technologies regardless of the costs involved, and to encompass an ever widening concept of health care that is, itself, responsive to the development of new technologies, the R & D sector will continue to face incentives that reward costly new measures relative to cost-reducing innovations. Such a reward system may not be incentive-compatible; new technologies may be developed even though they are welfare decreasing in the sense that the insured population is not willing to pay the real cost of developing and applying the technology (Goddeeris 1984b; Baumgardner 1989).

IV. Effects of R & D (Technological Change) on the Health Care Insurance System

Advances in medical technology---involving both diagnostics and treatmenthave been, at least arguably, a driving force behind the rapid growth of health care expenditures (Stuart Altman and Robert Blendon 1979; Jean Lacronique and Simone Sandier 1981; Jonathan Showstack, Stephen Shroeder, and Michael Matsumoto 1982; Henry Aaron and William Schwartz 1984; Wilensky 1987). The announcement for a recent (October 1988) conference cosponsored by the American Medical Association acknowledged the benefits from new medical technology but also cited the position that the growth of medical technology is a primary cause of the quadrupling of per capita health care costs between 1970 and 1986. Even if this causation occurs

however—and existing research is far from conclusive on the matter—the mechanism through which it works is not well understood. Neither is it apparent that technological advances would necessarily increase health care expenditures, rather than decrease them.

One mechanism through which technological change could foster increased expenditures on health care would be through its effect on the health care insurance system. If a previously untreatable condition becomes treatable, a possible outcome is that an individual could encounter a larger, but unpredictable, medical care expense for treatment than was previously the case; thus, both the mean and the variance of an individual's health care expenditures associated with that condition could increase.

Pooling of such risks is a logical response. In addition to the increased expected demand for private insurance, collective demand is also likely to increase; the fact that health care, particularly when it has a major effect on life expectancy or quality of life, is widely viewed as a "merit" good (or "altruistic externality"—Pauly 1986) results in public pressure on government to ensure that the care is available to whoever needs it medically, regardless of ability to pay.

An example of such a merit good is organ transplant technology. Reacting to the life-saving aspects of the new transplant technology, the Federal Government Task Force on Organ Transplantation recently proposed that government pay for all organ-transplant operations that patients cannot afford (Robert Pear 1986). Somewhat similar legislation, enacted in 1972-in response to the development of kidney dialysis (not transplant) technology-had the clear effect of increasing health care expenditures; no patient was rationed from access to the technology, and the technology, while life-extending, was more costly in resource terms (although not necessarily in net benefit terms) than simply allowing the victim to go without treatment and, hence, to die.¹⁴ The interplay of financial and political forces following the development of the dialysis technology (Richard Rettig 1980; Rettig and Ellen Marks 1983) and the massive public expenditures that ensued may help to explain why there has been no subsequent U.S. legislation covering such complete treatment for any other disease, and why the British National Health System continues to restrict access to dialysis for persons over age 55.

Life-extending technologies highlight the ambiguity of the concept of a technology being "expenditure increasing." Total health expenditures over a person's lifetime are likely to increase if the person lives longer, although that is not necessarily the case. However, expenditures per year of life can decrease even if lifetime expenditures increase. A new technology that increases the cost of treating a particular disease but is successful in increasing life expectancy sufficiently to decrease expected health care costs per year of life could diminish the demand for health care insurance; my conjecture is that it would not, but this deserves more attention.¹⁵ The point is that technological change need not increase demand for insurance, even if the change

¹⁴The view that dialysis and organ transplants are cost (or expenditure) increasing, ceteris paribus, deserves further comment as to what is embedded in the ceteris paribus assumption. One element is the set of probabilities of contracting all other diseases. The assumption that these probabilities are constant with respect to the organ transplant or dialysis decision may not be valid; a person whose life is "saved" through the use of one of these technologies may well face a greater probability of dying from other causes than do people who have not been victims of kidney disease.

¹⁵ The effect of increasing life expectancy on total health care expenditures as a percentage of GNP is yet another matter. This depends on the productivity of persons whose lives are extended, as well as on longer-run effects on birth rates. increases the expected cost of treating a particular illness. It could take forms that decrease either the aggregate expected health care cost for all illnesses, or the variance. Demand for insurance would also decline even if a new technology increased the aggregate expected cost of treatment, if the variance decreased sufficiently.¹⁶

11

1. 1. 1.

1. A. S. S. S.

. .

.....

٤

.

45.8

If we focus on treatment of specific diseases, we find that some innovations decrease the demand for insurance by decreasing both the expected cost of treating that illness, and the cost variance. The Salk and Sabin polio vaccines, for instance, are quite inexpensive to administer, and by providing immunity to the ravaging effects of polio, they have reduced-indeed, virtually eliminatedthe variance in health care expenditures associated with contracting that disease and using costly treatment technologies. The potentially enormous expenditures that have been eliminated, which include those associated with decades of use of an iron lung and the lifelong costs associated with being crippled, exceed the cost of providing the vaccinations (Weisbrod 1971). Thus, the polio vaccines, like many other vaccines, have the effect of reducing an individual's expected level of expenditures for treating the disease, as well as the variance around that mean. In the process they reduce the demand for health care insurance.

Organ transplant technology, on the

¹⁶ Even if technological change increases demand for insurance, it need not follow that the amount of insurance purchased would increase. Insofar as the technological changes were cost increasing, the price of insurance coverage would increase, which would diminish insurance purchases. In fact, the negative price effect of an increasing price for health care insurance appears not to have offset the positive demand-shift effect of technological change, judging from the growth in the fraction of the population with insurance; to be sure, however, much of the growth of insurance coverage over the last two decades has been through government rather than direct consumer purchases in private markets. other hand, is a technological advance that has increased both the mean and the variance of desired individual expenditures conditional on medical need. Before the new technology, a person with serious liver malfunction, for example, simply died, with comparatively little health care expenditure.¹⁷ With the new technology it has become possible to spend vast sums on effective treatment. A single liver transplant operation can cost \$200,000 or more, and subsequent medical attention and medication to prevent organ rejection typically totals \$10,000-\$20,000 annually for life (Jan Hudis 1986). Thus, a healthy person with some probability of developing liver disease faced a larger expected financial cost of treatment once the new technology was developed, and a greater variance in cost; conditional on remaining healthy, the person would spend zero on treatment of his or her liver under either technological state-with or without the transplant capability. Conditional on contracting liver disease, however, the person would spend a great deal more on treatment once the new technology became available. As a result, the development of transplant technology increased private demand for health care insurance, ceteris paribus. This is distinct from the increase in demand associated with the merit-good-related desire to provide access to life-saving technology to everyone regardless of ability to pay.¹⁸ These two cases of technological change-polio vaccines and organ trans-

¹⁷ In fact, however, little is known systematically about the amount of health care expenditures associated with attempts to cope with the debilitating effects of liver dysfunction (or other terminal illnesses), even when life is not prolonged.

¹⁸ Positive income effects associated with rising income could also account for an increase in the demand for health care insurance. One might expect, however, that the income elasticity would be negative, not positive; increased income, ceteris paribus, would increase the person's ability to self-insure (Jan Mossin 1968). Weisbrod: The Health Care Quadrilemma

plants-illustrate several points: (1) Some new technologies increase the expected health care expenditures for victims of a given disease, ceteris paribus, while others decrease it; (2) some new technologies increase the variance of health care expenditures for victims of a given disease, ceteris paribus, while others decrease it; (3) a technology that increases the mean and variance of health care expenditures for a particular disease would tend to increase the demand for health care insurance, while one that decreased them would tend to reduce the demand for insurance. This latter proposition suggests the following conjecture: The growth in insurance coverage, private and public, suggests that the preponderance of technological change in recent decades has increased the means and variances of health care expenditures associated with various diseases, rather than reduced them. Society has tended to develop a growing number of new technologies that permit higher levels of health care expenditures.¹⁹

Vaccines and transplants also illustrate stages in technical progress. Biologist Lewis Thomas (1975) distinguishes among three levels of technology in medicine: (1) "Nontechnology" tides patients over diseases that are poorly understood. It largely involves reassuring patients, providing hospitalization and nursing, but with little hope; "It is what physicians must do now for patients with intractable cancer, severe rheumatoid arthritis, multiple sclerosis, stroke, and advanced cirrhosis" (p. 37).

(2) At a higher level is "halfway technology." This includes dealing, after the

¹⁹Treatment of heart attacks is another illustration. One study showed that between 1972 and 1982, treatment of myocardial infarction involving more complex technologies such as cardiac imaging, angiography, and coronary bypass graft surgery was associated with a tripling of physician costs per case (Eric Sawitz et al. 1988). fact, with the incapacitating effects of diseases "whose course one is unable to do very much about." It is technology that adjusts to disease or postpones death. Examples include organ transplantations and artificial organs, and treatment of cancer through surgery, irradiation, and chemotherapy. The cancer measures are halfway technologies because they are directed at "already established cancer cells, but not at the mechanisms by which cells become neoplastic" (p. 39).

(3) "High technology," exemplified by immunization, antibiotics for bacterial infections, and by prevention of nutritional disorders, "comes as a result of a genuine understanding of disease mechanisms, and when it becomes available, it is relatively inexpensive . . . to deliver" (p. 40).

Thomas described the state of technology at a point in time-not the process of change. If, however, we think of a dynamic process, in which knowledge tends to grow from the first of the three levels to the second and then the third, the cost function associated with any particular disease might be inverted-U shaped; it is plausible, although certainly not verified, that health care costs are highest for the halfway technologies. In the extreme case of a nontechnology, when the knowledge base is so weak that there is nothing useful to be done, costs are likely to be low, as they are when the high technology state of knowledge is reached.

The evolution of knowledge about polio is a useful example. Two generations and more ago, the nontechnology stage prevailed. Many victims of the disease died quickly as a result of paralysis; for them, the effects were disastrous but the attendant health care costs were small. Development of the halfway (iron lung) technology prolonged life, but at substantial cost. The high technology polio vaccines (Sabin and Salk) dramatically reduced costs associated with polio, virtually eliminating it in the United States there were 5 cases in 1985, compared with over 38,000 in 1954, before the vaccines were developed.²⁰

Insofar as the inverted-U relationship holds between state of technology and resource cost per case, there is an interesting implication. The aggregate effect of technological change on health care costs will depend on the relative degree to which halfway technologies are replacing lower, less costly technologies, or are being replaced by new, higher technologies. The development of halfwav technologies was implicitly encouraged by the cost-reimbursement insurance system that has dominated hospital and medical care until recently, because there was little or no incentive for medical care providers to avoid costly technologies that were even marginally effective.²¹ Empirical research on how, and how much, the medical R & D process is now being affected by the shift to a prospective-pricing incentive system for cost control is in its infancy; there would seem to be an incentive for R & D to shift toward mechanisms that would bypass the high-cost, halfway states of technology.

Depending on whether technological change is predominantly from nontechnology to halfway, rather than from halfway to full or from nontechnology to full, the demand for insurance is likely to differ. With the demand for insurance being a function of uncertainty of loss, demand should tend to increase most rapidly

1

 20 Vaccines appear to be more cost reducing than they are. If vaccination cost is, say, \$5 per person, and if the incidence of the disease is one in 40,000, then the vaccine cost per case *prevented* is \$200,000. That may or may not be resource-cost saving, at least with respect to health care costs.

 21 "Halfway" technologies are not the only type of R & D encouraged by cost-based, retrospective insurance. Any technology with positive expected benefits is encouraged. when changes in technology are of the expenditure-increasing, halfway type. Costly new surgical techniques such as organ transplants and artificial replacement parts spur the demand for insurance; low-cost vaccines diminish it.²²

Why have there been relatively more developments of technologies like organ transplants than like the polio vaccines? Why, that is, has technological change in health care been "expenditure increasing"? Is it more than chance? To begin examining this issue, we turn to the effect of various kinds of insurance arrangements on incentives for the R & D sector to develop alternative types of technologies. For just as the forms of technological change affect the insurance system, so, too, does the insurance system affect the direction and pace of technological change. Depending on the type of insurance available to consumers, the R & D sector faces differing incentives to search for cost-reducing, "process" innovations relative to quality-increasing but cost-increasing, "product" innovations.

V. Effects of the Insurance/Finance System on R & D

Theory suggests the probable direction of the health care finance system's effects on R & D. Depending on hospitals' and physicians' incentives to adopt new technologies (which are contingent on the insurance system through which providers are paid) the R & D sector can face quite different financial incentives for both the level and direction of research. Fiscal pressure on health care providers to contain costs will affect the market for adoption of innovations, and by so doing, will alter R & D in predictable ways.

²² Thomas' typology applies to technologies used for prevention and treatment. While Thomas does not deal explicitly with technologies used for diagnosis,-we can think of those as complements to treatment; that is, costs of treatment include costs of determining which treatment mode to use.

The effects of insurance on R & D are not simply based on the existing insurance system, but on the system expected to exist in the future. The process of developing new medical technologies involves years of planning and research, and, when drugs and medical devices are involved, more years of clinical trials to obtain approval by the Food and Drug Administration; in the case of pharmaceuticals, a period lasting 12-15 years is typical between the initiation of a research process and the marketability of a drug. As a result of this lengthy process, the R & D process depends on forecasts of the health insurance system, for the form of expected insurance coverage will determine the strength of the market for new products. If, for example, decision makers in the R & D sector believed that development of a particular technology that was costly yet effective would cause government to expand insurance to cover it—as was done with kidney dialysis there could be an incentive to develop the product even though it was not covered under existing insurance.

By directing attention to the effect of health care insurance on R & D, I do not imply that insurance is the only force affecting R & D. Among other forces are the state of scientific knowledge, which affects the probability of scientific success from additional research; demographic variables, which affect the size of potential markets for new products; and political influences on the budget of the National Institutes of Health (NIH), which finances basic research. With respect to the NIH, it would be useful to learn more about the way the size and allocation of its scientific research budget are influenced, perhaps quite indirectly, by the health insurance system, through its impact on the eventual market for new technologies of various types.

Hospitals, physicians, and other health care providers select the resources used

to treat any particular patient within the technologically feasible set and subject to revenue constraints. These constraints depend partly on the insurance system, which influences both the diffusion of existing technologies and the expected profitability of potential new technologies (Joseph Newhouse 1981, 1988; Goddeeris 1987). Thus, the following proposition requires testing: The insurance/ finance system affects the incentives facing the R & D sector to develop new health care technologies of various "types." Because the demand confronting the health care R & D sector is derived from the demand facing health care providers, alternative insurance/finance systems will have differing long-run effects on the demand for innovations. In particular, insurance mechanisms can differ in the incentives they imply for reducing costs relative to enhancing quality.

The two types of insurance payment mechanisms — "retrospective," which pays a provider on the basis of "costs" incurred, and "prospective," which pays sums that are independent of those costs incurred²³—imply profoundly different incentives for both the development and diffusion of new technologies.

The claim that hospitals operate according to some "technological imperative" that determines medical choices (Victor Fuchs 1986) and that drives hospitals to adopt the latest technology, regardless of cost, may well have been correct, but the reason may have been less

²³ Arrow (1963) identifies three types of insurance, the third being "indemnity." This type, however, is a special case of prospective coverage in the sense that the insurer pays a fixed amount, conditional on a loss, but independent of the magnitude of the health care costs actually incurred. The indemnity might take the form of a fixed dollar payment for the loss of a limb, or for a given illness. If it took the form of a fixed dollar payment per day of hospitalization, it would have the character of retrospectivetype insurance.

mystical than the term suggests. The economic incentives explaining the "rapid and indiscriminate adoption of [medical] innovations" (Fuchs 1986, p. 29), and "the proclivity of doctors and hospitals to adopt almost any plausible new thing-drugs, surgical methods, equipment-that increases capability in any dimension . . . without regard to cost" (Richard Nelson 1972, p. 56), have been documented for such technologies as intensive care units, cobalt therapy, and the electroencephalograph (Louise Russell 1979). One "explanation" offered for the insensitivity to cost is an alleged lack of training of physicians and hospital administrators in weighing marginal benefits against marginal costs (Roger Battistella 1984). Even if this is valid, the impact of insurance-based incentives may well be powerful; "methods of third party payment . . . [do] not give [decision makers] any inducement to acquire that ability" (Fuchs 1986, p. 30).

Analyses of the effect of insurance on the adoption or diffusion of technologies have tended to concentrate on technologies that have already been developed. Less attention has been given to the implicit incentives for the R & D sector to develop various types of innovations. Retrospective pricing sends a clear signal to the R & D sector: Develop new technologies that enhance the quality of care, regardless of the effects on cost. Careful analysis remains to be done to distinguish causation from spurious correlation, but it appears that in the post-World War II era this signal produced the two results that could be expected: historically unequaled improvements in medical care technology-drugs, devices, diagnostics, and so on-and unprecedented growth in health care expenditures.²⁴

²⁴ Such increased costs might or might not pass a full benefit-cost test. The point, however, is that they contributed substantially to the accelerated growth of health care expenditures.

Transplantation of natural organs has already been mentioned as an example of a high-cost medical innovation made more likely by retrospective insurance. Another example is development of a wide range of implantable artificial joints and artificial organs. The human body has become increasingly like an automobile, with replacements available for an ever growing number of parts-an arm or a leg, at about \$2,000, an elbow at \$1,200, an ear at \$10,000, and a heart at \$50,000-\$80,000. They are even available in small, medium, large, and extra large sizes (N. R. Kleinfield 1983). "Installation," of course, is extra, and as with auto parts, is typically many times greater than the price of the part.

Technological advances in recent decades have given us spectacular innovations, but with scant attention to the resource costs of utilizing them. Open heart surgery can replace clogged arteries (coronary artery bypass graft surgery, CABG) but at a cost averaging \$46,000 (National Center for Health Services Research and Health Care Technology 1988). A baby born two and a half months prematurely and weighing well under two pounds can be kept alive, but at a cost of \$90,000 and with a 10 percent survival rate (Howard French 1989). Ultrasound technology, CT (computerized tomography) scanners, PET (positron emission tomography) scanners, and other diagnostic tools aid in disease detection but often at costs of tens of thousands of dollars per case detected-not counting the subsequent costs of surgery or other treatment. The PET scan, which aids in detecting heart disease at a cost of about \$1,800 per test-many times this for each case of heart disease detectedhas been argued to be only "slightly" better than the SPECT (single emission computed tomography) scan, which costs less than half (James Schiffman 1989). Under retrospective, cost-based financing, even small improvements have been adopted by physicians, hospitals, and other institutions which have had little or no incentive to balance social benefits against costs.

Consider, now, the reward structure implicit in an alternative insurance/finance system—prospective-payment, in which payment to a service provider is exogenous to provider decisions, conditional on admission of a patient. The particular version that is being applied to hospitals' Medicare patients, and increasingly to other patients as well, confronts a hospital (but not the patient's physician) with an exogenously determined set of prices, one for each of 468 diagnoses made at the time of admission.²⁵ No longer is gross revenue for treating a particular patient a function of the hospital's decisions on use of resources.

Financial incentives for hospitals under such a prospective payment arrangement differ diametrically from the incentives under retrospective payment. With a hospital's revenue being exogenous for a given patient once admitted, and an HMO's revenue being exogenous for a member for the given year, the organization's financial health depends on its ability to control costs of treatment.²⁶ Thus,

²⁵ In some instances diagnostic categories can be altered after admission, on the basis of information not available at admission. This produces some degree of revenue endogeneity, because the hospital and physician can decide on the amount of exploratory effort.

²⁶ While hospital revenue is largely exogenous once the patient is admitted, a hospital can influence both its gross and net (of cost) revenues through a variety of mechanisms for controlling admissions. A nongovernmental hospital may, for example, choose not to provide particular services such as an emergency room; it can decide which physicians may serve on its medical staff and, hence, which may admit patients; and it can provide its affiliated physicians with subtle but clear signals to "encourage" patients with complex problems to utilize governmental hospitals. Recent research is disclosing that, with the advent of prospective pricing for Medicare patients at most nongovernmental hospitals in 1983, there has been an increase in admissions to Veterans Administration under a prospective payment finance mechanism, the health care delivery system sends a vastly different signal to the R & D sector, with priorities the reverse of those under retrospective payment. The new signal is as follows: Develop new technologies that reduce costs, provided that quality does not suffer "too much." (The meaning of "too much" will be examined below.)

When a ceiling was placed on government payment for kidney dialysis, the direction of technical change was affected; large surface dialyzers were developed that cut the time required per session nearly in half, from 6–8 hours down to 3.5–4.5 hours. This led to substantial savings in professional labor costs, which are a major cost component (Rettig 1980).

The shift to a prospective-payment system (PPS) under Medicare appears to have brought about some of the expected changes in utilization of health services. PPS has not diminished use of intensive care units, but it has apparently decreased use of such diagnostic procedures as chest x rays; in the three years prior to PPS, 1980–83, the mean annual change in the number of chest x rays per Medicare patient discharge was zero, while for the 1983–85 period it decreased by 8 percent (Frank Sloan, Michael Morrisey, and Joseph Valvona 1988).

HMOs also present providers with an incentive to increase attention to costs relative to medical benefits. HMOs which are, in effect, mergers of health care providers and insurers—can be expected to adopt more slowly than would

(VA) hospitals, which are not included in the prospective, DRG pricing system; therefore we might expect them to receive more of the patients with illnesses likely to constitute financial "losers" to for-profit and voluntary nonprofit hospitals (Barbara Wolfe 1989). In the long run, when location of a hospital is variable, there is additional discretion for nongovernmental hospitals to locate in areas that are less likely to generate unprofitable cases.

a provider facing a retrospective pricing system, any new technology that is cost increasing, even if more effective. HMOs have been found to have lower costs per patient year than do nonmembers whose insurance was based on retrospective costs, largely attributable to a 30 percent lower rate of hospitalization (Harold Luft 1981); but the rate of introduction of new technologies does not appear to differ, at least as that is reflected in rates of change of per capita costs. The growth rate in total costs per person (including out-of-pocket costs) in the 1960s and 1970s appears to have been about the same (Newhouse et al. 1985) or "only slight ons in, and i making

some adjustments for selection bias.

The longer-term effects of PPS and HMOs on the R & D sector are more difficult to discern. There has been no formal modeling of the long-run effects on technical change of alternative payment systems for hospitals and physicians. Early literature attempting to explain the rising level of health care exper litures did not identify an imporhne' ical change. Subsetant r quenc ... times directed attention to the sect of technological change on health care costs (Altman and Blendon 1979), but that change in "qual-

ity and style of hospital care" was assumed implicitly to be exogenous---captured econometrically, perhaps, by a time trend (M. Feldstein 1971).

. .

The rate of diffusion of a number of existing technologies has been found to be responsive to insurance-related incentives (Russell 1979; Anthony Romeo, Judith Wagner, and Robert Lee 1984; Lee and Donald Waldman 1985; Sloan et al. 1986). There has been little study, however, of the effects of insurance on the R & D sector—private, governmental, and nonprofit—where new technologies are developed, although the linkage be-

tween the insurance system and incentives for the R & D sector has been noted (Paul Joskow 1981; Goddeeris -1984a, 1984b; Goddeeris and Weisbrod 1985; U.S. Congress 1985). The effect of prospective-payment insurance on R & D is illustrated by experience in the late 1980s with the cochlear implant for hearing-impaired persons; scientifically promising research was discontinued as a consequence of its expected unprofitability, which resulted from application of the DRG-pricing system. The 3 M Company, the manufacturer of the first FDAapproved single-channel cochlear implant model, halted research on a multichannel device because of hospitals' financial disincentives (Nancy Kane and Paul Manoukian 1989). Similarly, R & D on assistive communication devices for speech-impaired persons appears to have been retarded by the lack of insurance coverage; Medicare's payment policy favors inpatient over outpatient care, and there was "an administrative decision that the [communication] devices are not prosthetic devices needed for the functioning of a malformed body member" (U.S. Congress, Office of Technology Assessment 1984b, p. 30).

The current climate and incentives facing the R & D sector are not conducive to the development of costly new technologies. Another example is the newly emerging diagnostic procedure known as PET (positron emission tomography), "which produces three-dimensional images that reflect the metabolic and chemical activity of tissue" (see p. 536). PET is in clinical trial, but General Electric Company, its developer, "isn't making the kind of investment it did to rush CT (computerized tomography scanners) and MRI (magnetic resonance imaging devices) to market." According to a General Electric official, "The government is very cautious about approving reimbursement for PET. In the past, if a technology improved patient care, it would be approved. Now it must also be cost-effective" (Naj 1990, p. B4).

There are some further implications of the new incentives for hospitals to reduce costs rather than to increase quality. In the new era of prospective pricing of hospital services, we are likely to see a diversion of R & D resources away from new surgical techniques and toward lowercost substitutes, frequently pharmaceuticals. Surgical advances can be cost reducing, especially when they substitute for other halfway technologies; angioplasty, for example, substitutes for more costly coronary bypass graft surgery, and kidney transplantation substitutes for years of dialysis. When surgical advances substitute, however, for nontreatment, they are likely to increase the cost of treating the specific illness; because life expectancy may increase, though, the effect on mean annual health care costs per capita is less clear. Surgery is costly, relative to nonsurgical interventions, because it is labor-intensive, "custom" production-performed on a single patient; as such it has limited capacity for taking advantage of scale economies.²⁷ Increased use of surgery over the 1972-82 period, during which retrospective pricing of hospital services dominated, was the primary source of rising treatment costs for patients admitted to a teaching hospital for acute myocardial infarction, respiratory distress syndrome of the newborn, and other intensive treatments for the critically ill (Showstack, Mary Stone, and

²⁷Cost reductions are likely to result, however, from experience—learning-by-doing—which is a function of total accumulated volume, even if not a function of the rate of surgery per unit of time. In a study of six surgical procedures, including coronary artery bypass and hip replacement, between 1984 and 1986, it was found that mortality declined with volume for five of the six procedures, but current cost per case declined with volume for only two of the procedures. Data covered between 646 and 4,738 hospitals, depending on the procedure (Project Hope 1988). Shroeder 1985).²⁸ New surgical interventions are likely to be less attractive in a cost-containment environment.

By contrast with surgery, research on those pharmaceuticals that decrease expenditures, relative to those that increase quality but increase expenditures, is more attractive under prospective pricing. This is because demand patterns by hospitals (and HMOs) reflect the search for cost-reducing modes of treatment, including substitutes for costly surgical interventions; in particular, the advent of prospective pricing has increased the expected profitability of (a) R & D on drugs than can prevent the onset of costly treatments-vaccines, for example (John Huston and Weisbrod 1988) and of (b) R & D on drugs that substitute for surgery-for example, beta-blockers, which can substitute partially for coronary bypass surgery, and cimetidine, which substitutes for ulcer surgery (John Geweke and Weisbrod 1982).

Effects of PPS insurance on the pharmaceutical industry will not, however, be entirely favorable. Pharmaceuticals are not always substitutes for surgery; they are sometimes complements. Development of a new drug that complements surgery can increase the efficacy of surgery and thereby increase the demand for surgery—with major cost implications. In a cost-containment, insurance-finance environment, pharmaceutical industry R & D faces an incentive to develop drugs that substitute for surgery rather than enhance its efficacy.

Organ transplants illustrate the issue. Liver transplantation, a surgical technique, is effective today largely because of a recent technological advance in pharmaceuticals. The drug, cyclosporine, is

²⁸ Much of the medical literature reports findings for a single hospital. Whether the findings are generalizable to the entire höspital system is not clear.

crucial, because it suppresses the body's immune-system reaction to the transplanted organ; yet, unlike earlier immunosuppressant drugs, it does not stop the body from fighting off infections.

The good news about this technological breakthrough is that cyclosporine permits people with liver, kidney, and heart failure to be kept alive, living essentially normal lives. The bad news is that the resulting increase in the efficacy of organ transplant surgery has brought sharp increases in the usage of these very costly procedures. Only 26 liver transplants were performed in 1981, and, while increasing to nearly 1,200 by 1987, some 4,000-4,700 persons per year could benefit from the procedure. At a cost of about \$200,000 each, plus annual maintenance costs, meeting all the medical needs implies an annual cost of \$1 billion for this one procedure (National Organ Transplant Act 1983; Pear 1986; "Cyclosporine Turns Five" 1988). Heart, kidney, and other organ transplants suggest many times this level of potential expenditures as a consequence of the pharmaceutical breakthrough. It has also produced political pressure to ensure access to this lifesaving technology, regardless of a patient's ability to pay-pressure that is still being suppressed in part by the expedient of terming the procedures "experimental." The enormous expenditure potential of technological advances in drugs is currently highlighted by the drug AZT, which has been shown to be successful in prolonging life for patients with AIDS. The current estimate is that some 600,000 people can benefit from this drug, which, even with very recent reductions in price and dosage, costs about \$3,000 per year of treatment (Joann Lublin 1990), for a total potential cost of nearly \$2 billion.

Given the current financing environment, why are cost-increasing drugs such as AZT, cyclosporine, and TPA being de-

t

veloped? Cvclosporine came onto the market prior to the advent of PPS, and given the lengthy research and regulatory process in pharmaceutical research, it is reasonably clear that work related to AZT and TPA was well under way by the time prospective-type insurance incentives became powerful in the 1980s. Today, the fiscal pressures operating through Medicare, HMOs, and state Medicaid "formularies"-lists of drugs that will be paid for-are reducing drug company incentives to develop drugs for which "high" prices would be required to make the R & D effort profitable.

The form of insurance affects the direction of R & D not only in terms of quality relative to cost. It also affects the incentive to search for methods to treat the ill rather than to prevent their illness.²⁹ In general, health insurance has primarily covered treatment in hospitals, with preventive measures having quite limited coverage. As a result, the R & D sector has had less incentive to focus effort on prevention than on treatment, with the exception, perhaps, of vaccinations, for which government subsidization is common. Insofar as preventive measures are covered by insurance, they tend to involve technologies that utilize the "health care sector"-especially physicians and hospitals-even though other measures, such as better diet and exercise, might improve health at lower cost.³⁰

 29 In the long run, the price of private health insurance depends on the state of technology. Even so, risk-spreading over all the insured may make it privately profitable for the R & D sector to develop technologies for which the value (willingness to pay) is less than the social cost (Coddeeris 1984a, 1984b; Baumgardner 1989).

³⁰ Thus, prevention has its halfway technologies, too (as pointed out by an anonymous referee). For many forms of prevention, insurance is inappropriate because there is little uncertainty about the financial expenditure involved (thanks to Mark Satterthwaite for noting this).

I do not intend to imply that a reallocation of resources toward prevention would necessarily be efficient, given the existing state of knowledge (Russell 1986, 1987). Indeed, the concept of efficiency is itself controversial; it certainly can be defined in terms of either the patient's willingness to pay or some measure of health status, and in either private or social terms.³¹ The point is that today's state of knowledge about measures for preventing illness and for treating it reflects the historical incentives for R & D of both types, and those incentives have been shaped by the insurance system.

VI. Effects of Insurance on Choice of Technology and Quality of Health Care in the Short Run, with Technology Given

In addition to its potential to influence R & D, the health insurance incentive structure also influences the deployment of existing medical technology, with implications for quality and access to care. A retrospective, cost-based reward structure and a prospective reward structure such as a DRG system and an HMO³²

³¹ Aaron and Schwartz (1984) define efficiency in medical terms but using a Pareto-like approach: "Medical resources are efficiently used when a given total expenditure cannot be reallocated to alternative kinds of care to achieve an improved medical outcome. . . . [Thus] it would not be possible to increase total medical benefits by taking some money away from one service, for example cancer chemotherapy, and spending it on another, say x-ray" (pp. 79. 89). Randall Ellis and Thomas McGuire (1986) define efficient supply of care as existing when the physician acts as a perfect agent, weighing a dollar of hospital profit equally with a dollar of benefit to the patient.

³²There are important differences between an HMO- and a DRG-type payment system—at least as these operate now. For example, the DRG system applies currently only to hospital services, while HMOs cover a wider range of medical services. HMOs may operate their own hospitals, but they typically subcontract with independent hospitals for treatment of HMO members: such subcontracts can take many forms, and with either retrospective or prospective pricing. offer very different incentives for provider choice between increasing quality and decreasing costs (Michael Morrisey et al. 1984).

For a given state of technological knowledge, a prospective-payment insurance system provides encouragement, at the margin, to use production processes that reduce cost rather than improve quality. This is particularly so when quality is affected in dimensions that are costly for consumers (or regulators or insurers) to observe. The central point is that in a world of asymmetrically high information costs to consumers relative to service providers (e.g., hospitals and HMOs),33 it is harder to detect reductions in quality in some forms than in others, and the finance system can influence provider incentives to choose among input combinations that differ in the relative importance of effects that are more and less costly for nonproviders to monitor.

Every commodity—health care or anything else—can be thought of as a bundle of attributes that vary in the cost of monitoring them as well as in their importance to buyers. To simplify, consider two classes of attributes—type I, which is costless to monitor, and type II, which is costly to monitor. If consumers respond largely to the observable, type I attributes, then sellers will find price to be essentially independent of quality in the type II dimensions, and quality in the latter forms will be low (Weisbrod 1988). Price will be a poor gauge of overall quality.

A prospective-payment reward structure such as a DRG system is a price control mechanism. It poses the problem of how to ensure that real prices are not raised through the expedient of reducing

³³ An HMO, which vertically integrates a provider group with an insurer, reduces the informational asymmetry between the two, though not between either of them and consumer-patients or regulators. service quality, especially in the type II dimensions. 34

The potential effects of price setting by a governmental agency or private insurer when quality is asymmetrically costly to monitor can be seen by comparing the setting of prices for electricity and for care in a hospital or nursing home. A kilowatt-hour of electricity is far more homogeneous and easier to monitor than is a day of care (or any number of other potential measures of output) in a hospital or nursing home.³⁵ Thus, regulating price does not pose a serious risk that quality of output will be compromised in unobservable wavs by the regulatory process. Because of the more complex attributes of the health care system, opportunities are greater for providers to reduce output quality in dimensions that, being costly to monitor, are difficult to embody in a performance contract.

The Joint Commission on Accreditation of Health Care Organizations (JCAH) recognizes implicitly the distinction between type I and type II characteristics for assessing quality of a hospital. In his testimony at the 1973 Senate hearings, the executive director of JCAH said it was concerned with whether a hospital had the physical environment to permit high-quality medicine to be provided, for example, an operative sprinkler system (a type I attribute)—not with the actual clinical practices, for example, how carefully surgery is performed (a type II attri-

³⁴ Throughout this discussion the role of physicians as agents for patients has great importance. I assume that physicians act as imperfect agents, which leaves patients asymmetrically underinformed.

1

³⁵ There are other elements of the electric power regulatory process—for example, the "appropriate" level of inputs—that involve asymmetric costs. The literature on the Averch-Johnson effect focuses, in effect, on the difficulty regulators have in determining the degree of overcapitalization of public utilities under rate-of-return regulation (Harvey Averch and Leland Johnson 1962; William Baumol and Alvin Klevorick 1970). bute) (Kathleen Lohr, Karl Yordy, and Samuel Thier 1988).³⁶

I remarked earlier that under a prospective-payment system, financial incentives are to cut costs provided quality does not suffer "too much."37 There are consequences, of course, of cutting quality, and they constrain health care providers: tort law liability for medical malpractice, loss of patients to competitors (Albert Hirschman 1970), loss of donations and volunteer labor, and penalties for violating regulatory rules (Weisbrod and Mark Schlesinger 1986), professional ethics codes, and, in the case of HMOs, possibly greater costs of treating member-patients in the future.³⁸ Thus, the financial incentive to reduce costs by cutting quality is presumably equated at the margin with the effects of reduced quality on these revenue and cost variables (Robert Woodward and Frederick Warren-Boulton 1984; Ellis and McGuire 1986).³⁹ Little is known about the quantitative importance of each of these constraints, but because of them, a prospective-payment price control system implicitly encourages health care providers to cut resource use in the type II dimensions-which would minimize revenue losses and other

³⁶ John Porterfield, the JCAH executive director, reportedly said that a hospital reviewer would observe whether the hospital's sprinkler system worked and whether certain medical committees functioned and kept adequate records, but if a surgeon on the staff decided that good quality care required taking out the appendix of all blue-eyed males over age sixteen, that was none of the JCAH reviewers' business.

ness. ³⁷ Morrisey et al. (1984) model the effects on quality of care in a hospital confronted by downward price pressure.

³⁸ For HMOs the latter effect is attenuated by the uncertainty that the person will remain a member.

³⁹ Because HMOs involve a prospective payment to cover all "needed" care for the stipulated period, the incentives facing HMOs are analytically very similar to those facing hospitals under DRG pricing; thus; in general, propositions in this section referring to hospitals will also apply to HMOs, mutatis mutandis. penalties—not in ways that would be socially efficient. 40

Consumer-patients and donors cannot respond to changes in quality they cannot observe. Thus, given the imperfections in agency relationships (Ellis and McGuire 1986), the shift to a DRG-type prospective-payment insurance system can be expected to cause reductions in quality in precisely those forms that are difficult for insurers to monitor (Weisbrod 1989). This prediction requires testing, which needs to recognize that in a competitive market there can be simultaneous decreases in type II dimensions of quality and increases in type I dimensions. For example, increased "quality" in easily observed forms such as hospital candlelight dinners for maternity patients and spouses can attract patients to a hospital, and free dental or eye checkups can attract members to an HMO, even while quality of medical care is being reduced in more subtle, hard-to-detect forms (Weisbrod 1988, chs. 2, 3, 8).

The reuse of "disposable" items by hospitals illustrates the potential for cutting quality in ways that are difficult for consumers to monitor, and an effect of prospective pricing on the choice of production technology. Until the late 1940s. hospitals reused most medical devices; tubing, syringes, needles, and so on were made to be used, sterilized, and used again. When the new technology of disposables was introduced after World War II, it was quickly adopted by a health care finance system that encouraged the greater convenience and safety of disposables and deemphasized the cost consequences. The expanding system of retrospective-pay health insurance that covered all "reasonable" hospital costs spurred both the development and the

⁴⁰ This is analogous to "skimming" and "creaming" of program participants.

adoption of disposable items along with any other technology that was arguably quality enhancing.

Today, with the shift to prospective pricing, sterilization and reuse is returning. This change may or may not be efficient—allocatively or medically. What is striking is that hospitals are reusing items that are labeled by the manufacturers for "one-time-use only" (Alan Otten 1984). Even "disposable" filters for kidney dialysis machines are being reprocessed and reused (U.S. Congress, Office of Technology Assessment 1984a).

These practices reduce hospital costs. They may have no effect on revenues, for they are difficult for consumers (but presumably not their physician-agents) to observe. Thus, the financial consequences are relatively unambiguous. At the same time, the effect on health risks of reusing disposables is not currently known (Flora Chu et al. 1986; National Center for Health Services Research and Health Care Technology 1986). While the safety debate proceeds, the dispute is being resolved in favor of the costreducing technology. This is in sharp contrast to the situation in the 1950s, when the incentive structure was reversed; at that time, single-use disposables replaced the prior use-sterilize-reuse technology despite the absence of strong evidence of favorable health effects.

In general, the switch to prospective payment can be expected to bring changes in the technology of health care of just that type: they have clearly favorable effects on costs, but subtle or uncertain, yet presumptively nonpositive, effects on the quality of care. I say "presumptively" nonpositive because, given the state of technical knowledge, any change in resource use that is made after a change in incentives could have been made before; disposables could have been reused earlier.

Another quality-related dimension of hospital behavior likely to be affected by a shift to prospective pricing for hospitals is the length of a patient's stay. Confronted, under a DRG-pricing system, by a fixed price for treating each patient, hospitals have a financial incentive to discharge patients earlier (Judith Lave et al. 1988). Even if they do so, however, it is difficult for a patient to determine whether he or she has been discharged "quicker but sicker" (John Heinz 1986). Here, once again, a crucial question is how well asymmetrically underinformed patients are represented by physicianagents.

A reduction in use of hospital inputs is not necessarily inefficient in economic or medical terms; the cost saving may exceed the loss in benefits (although valuing the benefits is difficult), and in some situations there might be no medical benefits at all from, say, a longer hospital stay. Neither, though, is a reduction in inputs necessarily efficient. Input substitutions and cost reductions that may result from the shift from cost-based to prospective insurance cannot be assumed to be efficient or inefficient in a world of asymmetrically underinformed patientconsumers who confront prices that often bear little relationship to real marginal costs. Public policy, if it is to increase allocative efficiency, clearly demands understanding of the effects of pricing and other interventions on both quality and cost, not simply on costs. In particular, there should be attention to the tendency of a prospective payment insurance-pricing system to cause input substitutions that overvalue reductions in easily observed expenditures and undervalue reductions in quality that are more costly to observe.

1

The response of the health care sector to financial incentives may not be the same for its various institutional elements—private enterprise, governmen-

tal, and private nonprofit. In the hospital industry, 65 percent of all short-term beds are in private nonprofit hospitals, 26 percent in governmental. Thirty percent of nursing home beds are in nonprofit (22 percent) or government facilities (8 percent). Of kidney dialysis centers, 48 percent are nonprofit and an additional 12 percent are governmental (Weisbrod 1988). The key question is this: In response to a public policy shift from cost-based to prospective payment to providers, is there a different responsequantitatively or qualitatively-depending on the institutional ownership mix of the industry.⁴¹ Confronted by the incentives that prospective payment provides to discharge patients earlier and to engage in other forms of quality-shaving actions in the type II dimensions, do for-profit, nonprofit, and governmental organizations respond differently?42 Does institutional form matter?

Finding the answers to these questions requires modeling the behavior of each form of organization and the process of competition among them. There has been some attention to the conditions of equilibrium in institutionally mixed industries (Schiff 1986; Theodore Marmor, Mark Schlesinger, and R. W. Smithey 1986; Charles Phelps and Itai Sened 1989), but strong conclusions have not been reached.

Economic behavior may differ across ownership forms because of differences in objective functions, constraints, or

⁴²Whether earlier discharge of a hospital patient is a type I or type II attribute is debatable. I regard it as type II. While the length of stay for any patient is easily observed, what is difficult for the patient to observe is whether the length of stay was lower than it would have been if the physician and hospital were not responding to the altered financial incentive of PPS.

both. Profit maximization is typically assumed for the private enterprise components of the health care sector, but a variety of objective functions have been suggested for the nonprofit sector (Newhouse 1970; Karen Davis 1973; Pauly and Michael Redisch 1973; Estelle James 1983; Dennis Young 1983), as have various constraints on the distribution of profit⁴³ and access to public subsidies and private donations of money and time (Hansmann 1980; Susan Rose-Ackerman 1982; David Easley and Maureen O'Hara 1983; Alphonse Holtmann 1983; Charles Clotfelter 1985; Richard Steinberg 1986; Weisbrod and Nestor Dominguez 1986).44

DRG pricing provides the same financial incentive for all hospitals to discharge patients earlier than would a retrospective pricing system, but because of differences in objective functions and constraints, the behavioral responses may differ among institutional forms. There have been studies, for example, of the effect of prospective payment on

⁴³ Nonprofit organizations are not legally restricted from engaging in profitable activities; they are, however, restricted in what they may do with any profits. Little explicit attention has been devoted, however, to the enforceability of this constraint (Weisbrod 1988). This is relevant to the "managerial discretion" models of Oliver Williamson (1967), Armen Alchian and Harold Demsetz (1972), and Jean-Luc Migué and Gérard Bélanger (1974).

⁴⁴ All organizations, regardless of ownership, confront the same technological constraints, but they face different financial constraints in such forms as nonprofits' exemptions from property and sales taxes, and eligibility for postal subsidies. Charitable contributions of time and money to a nonprofit hospital (but not to a proprietary) might respond positively to the amount of unprofitable services it provides to low-income, uninsured, or other "deserving" persons. The relationship between donations to nonprofit organizations and the tax and expenditure behavior of government-the "crowding out" effecthas also received attention in the public finance literature. At the theoretic level see Peter Warr 1982; Russell Roberts 1984; Theodore Bergstrom, Laurence Blume, and Harold Varian 1986; James Andreoni 1988; for empirical studies see Burton Abrams and Mark Schmitz 1978, 1984; Schiff 1985.

the condition, at discharge, of elderly patients with hip fractures (Robert Palmer et al. 1989; John Fitzgerald, Patricia Moore, and Robert Dittus 1988) in two nonprofit hospitals, but they have not examined differences across ownership forms.⁴⁵

More generally, neither theory nor empirical tests have resolved the question of whether there are systematic differences among institutional forms. Econometric evidence, while mixed, is growing that when for-profit, nonprofit, and governmental organizations coexist in a given industry—as they do in hospitals and nursing homes, for examplethey do behave differently. Differences have been examined in four principal dimensions: (a) access to care, as reflected by admission of uninsured patients-that is, provision of "uncompensated" careand the use of waiting lists rather than prices, (b) quality of care, (c) cost efficiency, and (d) extent of opportunistic behavior toward asymmetrically underinformed consumers.

Systematic behavioral differences between private firms and nonprofit organizations have been found in some studies (Bradford Gray 1986, which summarizes a number of studies; Regina Herzlinger and William Krasker 1987; Lawrence Lewin, Timothy Eckels, and Linda Miller 1988; Weisbrod 1988; Thomas Selden 1989), but not in others (Robert Clark 1980; Sloan and Robert Vraciu 1983; Gary Gaumer 1986). Nonprofit providers of health care, especially the church-affiliated nonprofits, appear to utilize a somewhat greater proportion of their resources to care for the indigent,

⁴⁵ Palmer et al. (1989) found no change in ambulation status, comparing patients discharged from one nonprofit hospital in the several years before and after the change in price incentives. Fitzgerald et al. (1988), studying a single "community" hospital (presumably also a nonprofit), found significantly reduced mobility.

⁴¹ A related issue is how competition among organizations of various ownership types affects long-run equilibrium, and whether one form of institution can be expected to drive out the others (Jerald Schiff and Weisbrod 1987).

they provide a wider range of services (and in this sense, higher quality), and they take less advantage of their informational advantages over patients.

Neither the underlying theory nor the available, nonexperimental data, however, are yet strong enough to justify confident generalizations about differences in institutional behavior. Measuring quality of service in a hospital (Stephen Shortell and Edward Hughes 1988), controlling for differences in patient conditions, and distinguishing care of the indigent from "bad debts" associated with poor management, all remain subjects for future research, as does any differential responsiveness to the development of new technologies.⁴⁶ There is also a question of the appropriate estimation modeling; many econometric efforts to detect differential behavior across institutional forms may have misspecified their models, controlling erroneously for variables such as organization size, which are endogenous to the choice of institutional form (Weisbrod and Elizabeth Mauser 1990).

VII. Concluding Remarks

Economists' concerns about skyrocketing health care expenditures have focused heavily on insurance and its encouragement of inefficiently great utilization. Yet it is clear that much of the growth in health care expenditures during the post-World War II period has resulted not from increased prices for existing technologies, but from the price for new technologies. Newly developed technologies have driven up both costs of care and the demand for insurance, while also expanding the range of services for which consumers demand insur-

.

⁴⁶ In a related study of rapidity of introduction of new technologies in HMOs relative to fee-for-service providers, the Rand Corporation health insurance experiment found an apparently slower rate of introduction in HMOs (Newhouse et al. 1985).

ance. At the same time, expanding insurance coverage, which includes more people as well as a growing array of health care inputs, has provided an increased incentive to the R & D sector to develop new technologies, and a growing incentive for subsets of consumers who could benefit from particular new technologies to seek a wider definition of what would be covered by insurance. Both the resource costs of health care and our technical ability to prolong life and enhance its quality have risen sharply. The interactive process involving insurance and R & D is still evolving. It is increasingly being influenced by the recent change in incentives associated with the shift from retrospective, cost-based insurance coverage to prospective, exogenously determined pricing.

Although this paper has focused on the health care sector, the kinds of incentive effects it has examined are quite general. As an example of the potential effect of insurance on incentives facing the R & D sector, consider another major area of public policy and expenditure-education. Unlike health care, which has been financed for decades by a retrospective, cost-based finance system, elementary and secondary education has been financed traditionally through what amounts to a prospective payment system; roughly speaking, state and local governments have given the schools a fixed grant per child. This is roughly analogous to a DRG system with a single DRG, so that every patient (child) entering a hospital (school) brings a fixed sum of revenue to the provider. A school district can also be thought of as, like an HMO, providing "comprehensive" services to all "members" (students) in return for a fixed annual fee. By examining how the interaction of finance mechanisms and R & D incentives have operated in the health and education areas, we can gain insight into what the health care system would be like today had we

taken an alternate route for financing it, as well as how a change in school finance would be likely to affect the education system.

Assume that public schools had been financed differently—in the way hospitals have been financed until recently: (1) school revenue was determined through a retrospective (cost-based) pricing system, in which (2) teachers were empowered to decide what resources should be used (a) to diagnose a particular child's educational "needs" and (b) to meet those needs, and (3) a bill for the cost of the resources used for each child was sent to government or a private insurer and subsequently paid to the school district.

Two questions arise: If such a system had been adopted after World War II for schools, what would have happened over the subsequent 40 years to the level of education expenditures? What would have happened to the pace of technological change in education? The lessons from health care suggest conjectures: If schooling had been "insured" on the basis of retrospective costs, expenditures would have increased far more rapidly than they did; and the pace of technological innovation in schools would have been far greater than it was.

Because education actually utilized a prospective pricing system, while health care utilized a retrospective pricing system, it is interesting to compare the two programs in terms of expenditure growth and technological change. First, with respect to expenditures, the share of GNP devoted to public elementary and secondary education has changed little over several decades (in which enrollments have remained relatively constant); between 1960 and 1985, for example, years of virtually identical enrollments-36.7 million and 36.6 million-public school expenditures increased from 3.03 percent of GNP to 3.42 percent (U.S. Bureau of the Census 1987 tables 186, 100

and 698); meanwhile, health care expenditures were rising from 4.6 percent of GNP to 10.7 percent (U.S. Bureau of the Census 1975, 1987).

Second, with respect to the pace and nature of technological change that might have occurred in education had retrospective pricing prevailed, we can do some informed speculating. To begin, we can compare-impressionistically-the technological change that has occurred in health care and in education. The typical hospital, for example, is barely comparable to its counterpart several decades ago, with entirely new techniques and facilities for diagnosis and treatment. The typical school, however, differs far less from its post-World War II counterpart, utilizing similar classrooms, teachers trained in similar ways, and using instructional techniques that, despite some computerization in recent years, employ capital-labor ratios that have changed relatively little.

We can predict that if retrospective reimbursement had prevailed for schools, the private sector would have devoted more resources to development of "improved" educational diagnostic and learning technologies; had that been the case, we would probably find now that education, like health care, had improved dramatically, but that society was paying a great deal more for it.

Today, the public policy "problems" in health care and in education are perceived to be sharply different, and in ways that correspond to the differences in finance mechanisms (although other forces are doubtless also at work). In health care, the central policy focus is on control of expenditures, quality of care not being seen generally as a problem.⁴⁷ In education, it is the reverse—the policy focus is on "low" quality of education,

⁴⁷ Problems of the uninsured are serious, but are receiving less attention than is the general problem

control of school expenditures receiving relatively less attention.

The ideas presented above are a mixture of solid knowledge, soft knowledge, and hypotheses requiring testing. In order to expand knowledge about health care and provide financial access to it, we need to understand more fully the dvnamic process through which the health insurance sector, private and public, interacts with the R & D sector. This area offers a rich research agenda with enormous potential, for the policy implications extend far beyond health care and across geographical boundaries.

References

- AARON, HENRY J. AND SCHWARTZ, WILLIAM B. The painful prescription: Rationing her h care. Washington, DC: Brookings Institutior 19-34.
- ABRAMS, BURTON A. AND SCHMITZ, MARK D. "The 'Crowding-Out' Effect of Governmental Transfers on Private Charitable Contributions," Public Choice, 1978, 33(1), pp. 29-39.
- "The Crowding-Out Effect of Governmental Transfers on Private Charitable Contributions: Cross-Section Evidence," Nat. Tax J., Dec. 1984, 37(4), pp. 563-68.
- AKERLOF, GEORGE. "The Market for 'Lemons': Qualitative Uncertainty and the Market Mechanism, Quart. J. Econ., Aug. 1970, 84(3), pp. 488-500. ALCHIAN, ARMEN AND DEMSETZ, HAROLD. "Production, Information Costs, and Economic Organization," Amer. Econ. Rev., Dec. 1972, 62(5), pp. 777-95.
- ALTMAN, STUART H. AND BLENDON, ROBERT. eds. Medical technology: The culprit behind health care costs? Washington, DC: U.S. GPO, 1979. DHEW Publication no. PHS 79-3216.
- ANDREONI, JAMES. "Privately Provided Public Goods in a Large Economy: The Limits of Altruism," J. Public Econ., Feb. 1988, 35(1), pp. 57-73.
- ABROW, KENNETH. "Uncertainty and the Welfare Economics of Medical Care," Amer. Econ. Rev., Dec. 1963. 53(5), pp. 941-73.
- AVERCH, HARVEY AND JOHNSON, LELAND, "Behavior of the Firm under Regulatory Constraint," Amer. Econ. Rev., Dec. 1962, 52(5), pp. 1052-69.
- BATTISTELLA, ROGER. "Review of The Public/Private Mix for Health, by Gordon McLachlan and Alan MAYNARD, eds.," Inquiry, Spring 1984, 21(1), pp. 98-99.
- BAUMGARDNER, JAMES R. "Forms of Insurance and the Pattern of Aggregate Technical Change in Medical Care." Working Paper, Duke U. Dept. of Economics, 1989.
- BAUMOL, WILLIAM J. AND KLEVORICK, ALVIN K. "Input Choices and Rate-of-Return Regulation: An

Overview of the Discussion," Bell Yearbook 1. NY and London: Praeger, 1970, pp. 162–90.

- BERGSTROM, THEODORE; BLUME, LAURENCE AND VARIAN, HAROLD. "On the Private Provision of Public Goods," J. Public Econ., Feb. 1986, 29(1), pp. 25-49.
- BOBEN, PAUL. "Modeling the Shift from Retrospective to Prospective Pricing of Health Care." Discussion paper, U. of Wisconsin-Madison, Center for Health Economics and Law, 1989.
- CHERNICK, HOWARD A.; HOLMER, MARTIN R. AND WEINBERG, DANIEL H. "Tax Policy Toward Health Insurance and the Demand for Medical Services, J. Health Econ., Mar. 1987, 6(1), pp. 1-25.
- CHU, FLORA ET AL. "Reuse and Reprocessing of Disposable Medical Devices: Legal Liability Issues," J. Health Care Technology, Summer 1986, 3(1), pp. 5-12.
- CLARK, ROBERT C. "Does the Nonprofit Form Fit the Hospital Industry?" Harvard Law Review, May 1980, 93(7), pp. 1416-89.
- CLEETON, DAVID; GOEPFRICH, VALY T. AND WEIS-BROD, BURTON A. "The Consumer Price Index for Prescription Drugs: Dealing with Technological Change." Working Paper, Northwestern U., Center for Urban Affairs and Policy Research, 1990.
- CLOTFELTER, CHARLES T. Federal tax policy and charitable giving. Chicago: U. of Chicago Press, 1985.
- CROMWELL, JERRY AND MITCHELL, JANET B. "Physician-Induced Demand for Surgery," J. Health Econ., Dec. 1986, 5(4), pp. 293-313.
- Cyclosporine Turns Five," Science, Oct. 14, 1988, 242(4876), p. 198.
- DANZON, PATRICIA M. Medical malpractice. Cambridge: Harvard U. Press, 1985.
- DAVIS, KAREN. "Theories of Hospital Inflation: Some Empirical Evidence," J. Human Res. Spring 1973, 8(2), pp. 181–201.
- EASLEY, DAVID AND O'HARA, MAUREEN. "The Economic Role of the Nonprofit Firm," Bell J. Econ., Autumn 1983, 14(2), pp. 531-38.
- "Elective Surgery: Cut it Out," Science News, Jan. 6, 1979, 115(1), p. 9.
- ELLIS, RANDALL P. AND MCGUIRE, THOMAS G. "Provider Behavior Under Prospective Reimbursement," J. Health Econ., June 1986, 5(2), pp. 129-51.
- EVANS, ROBERT G. "Supplier-Induced Demand: Some Empirical Evidence and Implications," in The economics of health and medical care. Ed.: MARK PERLMAN. London: Macmillan, 1974, pp. 162-73.
- FELDSTEIN, MARTIN S. "Hospital Cost Inflation: A Study of Nonprofit Dynamics," Amer. Econ. Rev., Dec. 1971, 61(5), pp. 853-72.
- FELDSTEIN, MARTIN S. AND ALLISON, ELIZABETH. "Tax Subsidies of Private Health Insurance," in The economics of federal subsidy programs. Part 8. Papers submitted to the Subcommittee on Priorities and Economy in Government, Joint Economic Committee, 93rd Congress, 2nd session, July 1974.

FELDSTEIN, MARTIN S. AND FRIEDMAN, BERNARD. "Tax Subsidies, the Rational Demand for Insur-

ance, and the Health Care Crisis," J. Public Econ., Apr. 1977, 7(2), pp. 155-78. FELDSTEIN, PAUL J. The politics of health legislation: 49. An economic perspective. Ann Arbor, MI: Health Administration Press, 1988. FITZGERALD, JOHN F.; MOORE, PATRICIA S. AND DIT-TUS, ROBERT S. "The Care of Elderly Patients with Hip Fracture: Changes Since Implementation of the Prospective Payment System," New England J. Medicine, Nov. 24, 1988, 319(21), pp. 1392-97. FRENCH, HOWARD W. "Tiny Miracles Become Huge Public Health Problem," New York Times, Feb. 350-65. 19, 1989, pp. 1, 22. FUCHS, VICTOR R. The health economy. Cambridge: Harvard U. Press, 1986. 1981. GARC, MOHAN L.; GLIEBE, WERNER A. AND ELKHA-TIB, MOUNIR B. "The Extent of Defensive Medicine: Some Empirical Evidence," Legal Aspects of Medical Practice, Feb. 1978, 6(1), pp. 25-GAUMER, GARY L. "Medicare Patient Outcomes and Hospital Organizational Mission," in For-profit enterprise in health care. Ed.: BRADFORD H. GRAY. Washington, DC: National Academy Press, 1986, pp. 354-74. GEWEKE, JOHN AND WEISBROD, BURTON A. "Clinical Evaluation vs Economic Evaluation-The Case of a New Drug," Medical Care, Aug. 1982, 20(8), pp. 821-30. GODDEERIS, JOHN H. "Insurance and Incentives for Innovation in Medical Care," Southern Econ. J. Oct. 1984a, 51(2), pp. 530-39. "Medical Insurance, Technological Change, and Welfare," Econ. Inquiry, Jan. 1984b, 22(1), pp. 56-67. "Economic Forces on Hospital Technology: Lessons from the U.S. Experience," International J. Technology Assessment in Health Care, 1987, 3(2), pp. 223-40. GODDEERIS, JOHN H. AND WEISBROD, BURTON A. "What We Don't Know About Why Health Care Expenditures Have Soared: Interaction of Insurance and Technology," Mount Sinai J. Medicine, Nov. 1985, 52(9), pp. 685-91. GRAY, BRADFORD, H. ed. For-profit enterprise in health care. Washington, DC: National Academy of Science Press, 1986. HANSMANN, HENRY. "The Role of Non-profit Enterprise," Yale Law J., Apr. 1980, 89(5), pp. 835-HEINZ, JOHN. "The Effects of DRG's on Patients," Business and Health, July/Aug. 1986, 3(8), pp. 17-HERZLINGER, REGINA AND KRASKER, WILLIAM "Who Profits from Nonprofits," Harvard Bus. Rev., Jan/ Feb. 1987, 65(1), pp. 93-106. 1981. HIRSCHMAN, ALBERT O. Exit, voice, and loyalty. Cambridge: Harvard U. Press, 1970. HOGAN, C. Urban and rural hospital costs: 1981-85. DHHS Publication (PHS) 88-3419 (Apr. 1988), Hospital Studies Program Research Note 12, National Center for Health Services Research and

29.

901.

20.

Health Care Technology Assessment. Rockville,

MD: Public Health Service, 1988.

HOLTMANN, ALPHONSE G. "A Theory of Non-Profit Firms," Economica, Nov. 1983, 50(200), pp. 439-

- HUDIS, JAN. "The Organ Game," Health, Oct. 1986, 18(10), pp. 61-66ff.
- HUSTON, JOHN AND WEISBROD, BURTON A. "Benefits and Costs of Human Vaccines." Working Paper, U. of Wisconsin-Madison, Center for Health Economics and Law, 1988.
- JAMES, ESTELLE. "How Nonprofits Grow: A Model," J. Policy Anal. Manage., Spring 1983, 2(3), pp.
- JOSKOW, PAUL. Controlling hospital costs: The role of government regulation. Cambridge: MIT Press,
- KANE, NANCY M. AND MANOUKIAN, PAUL D. "The Effect of the Medicare Prospective Pavment System on the Adoption of New Technology: The Case of Cochlear Implants," New England J. Medicine, Nov. 16, 1989, 321(20), pp. 1378-83.
- KLEINFIELD, N. R. "Companies in Search of Bionic Man," New York Times, Nov. 20, 1983, sect. 3, pp. 1, 26-27.
- LACRONIQUE, JEAN F. AND SANDIER, SIMONE. "Technological Innovation: Cause or Effect of Increasing Health Expenditures," in Innovation in health policy and service delivery. Ed.: CHRISTA ALTENSTET-TER. Cambridge: Oelgeschlaher, Gunn & Hain, 1981, pp. 87-105.
- LAVE, JUDITH R. ET AL. "The Early Effects of Medicare's Prospective Payment System on Psychiatry," Inquiry, Fall 1988, 25(3), pp. 354-63.
- LEE, ROBERT H. AND WALDMAN, DONALD M. "The Diffusion of Innovations in Hospitals,"J. Health Econ., Dec. 1985, 4(4), pp. 373-80.
- LETSCH, SUZANNE, W.; LEVIT, KATHERINE R. AND WALDO, DANIEL R. "National Health Expenditures, 1987," Health Care Financing Review, Winter 1988, 10(2), pp. 109-22.
- LEWIN, LAWRENCE S.; ECKELS, TIMOTHY J. AND MILLER, LINDA B. "Setting the Record Straight: The Provision of Uncompensated Care by Not-for-Profit Hospitals," New England J. Medicine, May 5, 1988, 318(18), pp. 1212-15.
- LOHR, KATHLEEN N.; YORDY, KARL D. AND THIER, SAMUEL O. "Current Issues in Quality of Care," Health Affairs, Spring 1988, 7(1), pp. 5-18.
- LUBLIN, JOANN S. "FDA Lets Wellcome Cut Dosage of Drug for AIDS, Easing Pressure for Price Cut, Wall Street Journal, Jan. 17, 1990, p. B2.
- LUFT, HAROLD S. "Trends in Medical Care Costs: Do HMOs Lower the Rate of Growth?" Medical Care, Jan. 1980, 18(1), pp. 1-16.
- _. Health maintenance organizations: Dimensions of performance. NY: John Wiley & Sons,
- MANNING, WILLARD G. ET AL. "Health Insurance and the Demand for Medical Care: Evidence from a Randomized Experiment," Amer. Econ. Rev., Jun. 1987, 77(3), pp. 251-77.
- MARMOR, THEODORE; SCHLESINGER, MARK AND SMITHEY, R. W. "A New Look at Nonprofits: Health Care Policy in a Competitive Age," Yale J. Regulation, Spring 1986, 3(1), pp. 313-49.

- MIGUÉ, JEAN-LUC AND BÉLANGER, GÉRARD. "Toward a General Theory of Managerial Discretion," Public Choice, Spring 1974, 17(4), pp. 27–47.
- MITCHELL, BRIDGER M. AND PHELPS, CHARLES E. "National Health Insurance: Some Costs and Effects of Mandated Employee Coverage," *J. Polit. Econ.*, June 1976, 84(3), pp. 553–71.
- MITCHELL, BRIDGER M. AND VOGEL. RONALD J. "Health and Taxes: An Assessment of the Medical Deduction," Southern Econ. J., Apr. 1975, 41(4), pp. 660-72.
- MORRISEY, MICHAEL A., ET AL. "Hospital Rate Review: A Theory and an Empirical Review," J. Health Econ., Apr. 1984, 3(1), pp. 25–47.
- MOSSIN, JAN. "Aspects of Rational Insurance Purchasing," J. Polit. Econ., July/Aug. 1968, 76(4), pp. 553-68.
- NAJ, AMAL KUMAR. "Diagnostic Equipment Field in Squeeze," Wall Street Journal, Sept. 6, 1990, pp. B1, B4.
- NATIONAL CENTER FOR HEALTH SERVICES RESEARCH AND HEALTH CARE TECHNOLOGY. The reuse of hemodialysis devices labeled "For Single Use Only." Washington, DC: National Center for Health Services Research and Health Care Technology, 1986.
- National Organ Transplant Act. Hearings before the Subcommittee on Health and the Environment, July 29, Oct. 17, 31, 1983. Washington, DC: U.S. GPO, 1983.
- NAZARIO, SONIA L. "Infertility Insurance Gains Backing," Wall Street Journal, Dec. 5, 1989, pp. B1, B5.
- NELSON, RICHARD R. "Issues and Suggestions for the Study of Industrial Organizations in a Regime of Rapid Technical Change," in *Policy issues and re*search opportunities in industrial organization. Ed.: VICTOR R. FUCHS. NY: National Bureau of Economics Research and Columbia U. Press, 1972. NEWHOUSE, JOSEPH. "Toward a Theory of Nonprofit Institutions: An Economic Model of a Hospital."
- Amer. Econ. Rev., Mar. 1970, 60(1), pp. 64-75. ————. "Erosion of the Medical Marketplace," Advances in Health Economics and Health Services Research, 1981, 2, pp. 1-34.
- "Has the Erosion of the Medical Marketplace Ended?" J. Health, Politics, Policy, and Law, Summer 1988, 13(2), pp. 263–79.
- NEWHOUSE, JOSEPH P. ET AL. "Are Fee-for-Service Costs Increasing Faster Than HMO Costs?" Medical Care, Aug. 1985, 23(8), pp. 960–66.
- ORGANIZATION FOR ECONOMIC CO-OPERATION AND DE-VELOPMENT. Health Care Systems in Transition. Paris: OECD, 1990.
- OTTEN, ALAN L. "Reuse of Devices by Hospitals and Doctors Stirs Debate Over Cost Savings and Safety," Wall Street Journal, Sept. 14, 1984, p. 27.

.

PALMER, ROBERT M. ET AL. "The Impact of the Pro-

spective Payment System on the Treatment of Hip Fractures in the Elderly," Archives of Internal Medicine, Oct. 1989, 149(10), pp. 2237-14.

PAULY, MARK V. "Overinsurance and Public Provision of Insurance: The Roles of Moral Hazard and Adverse Selection," Quart. J. Econ., Feb. 1974, 88(1), pp. 44-62.

Taxation. Health Insurance, and Market Failure in the Medical Economy," J. Econ. Lit., June 1986, 24(2), pp. 629–75.

PAULY, MARK V. AND REDISCH, MICHAEL. "The Not-For-Profit Hospital as a Physicians' Cooperative," Amer. Econ. Rev., Mar. 1973, 63(1), pp. 87– 100.

PEAR, ROBERT. "Federal Payment for Transplants for Poor Studied," New York Times. May 18, 1986, sect. 1, p. 1, 31.

- PHELPS. CHARLES E. AND SENED, ITAI. "Market Equilibria with Not-For-Profit Firms." Working Paper 191, U. of Rochester, Dept. of Political Science, June 1989.
- PROJECT HOPE. Trends in the Concentration of Six Surgical Procedures Under PPS and Their Implications for Patient Mortality and Medicare Costs. Chevy Chase, MD: Project Hope, Center for Health Affairs, 1988.

REINHARDT, Uwe. "The Theory of Physician-Induced Demand: Reflections After a Decade," J. Health Econ., June 1985, 4(2), pp. 187-93.

Report of the President's Biomedical Research Panel. Washington, DC: U.S. GPO, 1976.

RETTIC, RICHARD A. "The Politics of Health Cost Containment: End-Stage Renal Disease," Bulletin of the New York Academy of Medicine, Jan./Feb. 1980, 56(1), pp. 115-38.

RETTIG, RICHARD A. AND MARKS, ELLEN. The federal government and social planning for end-stage renal disease: Past, present, and future. Santa Monica, CA: Rand Corporation, 1983.

ROBERTS, RUSSELL D. "A Positive Model of Private Charity and Public Transfers," J. Polit. Econ., Feb. 1984, 92(1), pp. 136–48.

- ROMEO, ANTHONY A.; WAGNER, JUDITH L. AND LEE, ROBERT H. "Prospective Reimbursement and the Diffusion of New Technologies in Hospitals, J. Health Econ., Apr. 1984, 3(1), pp. 1–24.
- ROSE-ACKERMAN, SUSAN. "Charitable Giving and 'Excessive' Fundraising," Quart. J. Econ., May 1982, 97(2), pp. 193-212.

ROSSITER, LOUIS F. AND WILENSKY, GAIL R. "Identification of Physician-Induced Demand," J. Human Res., Spring 1984, 19(2), pp. 231-44.

RUSSELL, LOUISE B. Technology in hospitals. Washington, DC: Brookings Institution, 1979.

Is prevention better than cure? Washington, DC: Brookings Institution, 1986.

- SAWITZ, ERIC, ET AL. "The Use of In-Hospital Physician Services for Acute Myocardial Infarction

Changes in Volume and Complexity Over Time," J. American Medical Association, Apr. 1988, 259(16), pp. 2419-22.

- SCHIFF, JERALD. "Does Government Spending Crowd Out Charitable Contributions?" National Tax J., Dec. 1985, 38(4), 535–46.
- SCHIFF, JERALD AND WEISBROD, BURTON A. "Competition Between For-Profit and Nonprofit Organizations." Working Paper, U. of Wisconsin-Madison, Dept. of Economics, Dec. 1987.
- SCHIFFMAN, JAMES R. "Debate Grows Over Clinics Pushing Costly PET Scans," Wall Street Journal, Apr. 3, 1989, p. B1.
- SELDEN, THOMAS M. "Does Institutional Form Matter? Economic Behavior in the Inpatient Mental Health Care Industry." Research Paper 6, U. of Wisconsin-Madison, Mental Health Research Center, July 1989.
- SHORTELL. STEPHEN M. AND HUGHES, EDWARD F. X. "The Effects of Regulation, Competition, and Ownership on Mortality Rates Among Hospital Inpatients," New England J. Medicine, Apr. 28, 1988, 318(17), pp. 1100-06.

SHOWSTACK, JONATHAN A.; SHROEDER, STEPHEN A. AND MATSUMOTO, MICHAEL F. "Changes in the Use of Medical Technologies, 1972–1977: A Study of 10 Inpatient Diagnoses," New England J. Medicine, Mar. 25, 1982, 306(12), pp. 706–12.

- SHOWSTACK, JONATHAN A.; STONE, MARY HUGHES AND SHROEDER, STEVEN A. "The Role of Changing Clinical Practices in the Rising Costs of Hospital Care," New England J. Medicine, Nov. 7, 1985, 313(19), pp. 1201-07.
- SLOAN, FRANK A.; MORRISEY, MICHAEL A. AND VAL-VONA. JOSEPH. "Medicare Prospective Payment and the Use of Medical Technologies in Hospitals." Medical Care, Sept. 1988, 26(9), pp. 837– 53.
- SLOAN, FRANK A. ET AL. "Diffusion of Surgical Technology: An Exploratory Study," J. Health Econ., Mar. 1986, 5(1), pp. 31-62.
- SLOAN, FRANK A. AND VRACIU, ROBERT A. "Investor-Owned and Not-For-Profit Hospitals: Addressing Some Issues," *Health Affairs*, Spring 1983, 2(1), pp. 25-37.
- STANO, MIRON. "A Clarification of Theories and Evidence on Supplier-Induced Demand for Physician Services," J. Human Res., Fall 1987, 22(4), pp. 611-20.
- STEINBERG, RICHARD. "Should Donors Care about Fundraising?" in The economics of nonprofit institutions. Ed.: SUSAN ROSE-ACKERMAN. NY: Oxford U. Press, 1986, pp. 347-64.
- STEVENS, ROSEMARY. In sickness and in wealth: American hospitals in the twentieth century. NY: Basic Books, 1989.
- TAYLOR, AMY K. AND WILENSKY, GAIL R. "The Effect of Tax Policies on Expenditures for Private Health Insurance." in Market reforms in health care. Ed.: JACK A. MEYER. Washington, DC: American Enterprise Institute 1983 np. 162-84

THOMAS, LEWIS. The lives of a cell. NY: Bantam Books, 1975.

- TITMUSS, RICHARD M. The gift relationship: From human blood to social policy. NY: Pantheon Books, 1971.
- U.S. BUREAU OF THE CENSUS. Statistical Abstract of the United States, 1975. Washington, DC: U.S. GPO. 1975.
- _____. Statistical Abstract of the United States, 1987. Washington, DC: U.S. GPO, 1987.
- _____. Statistical Abstract of the United States, 1989. Washington, DC: U.S. GPO, 1989.
- U.S. CONGRESS, OFFICE OF TECHNOLOGY ASSESS-MENT. The hemodialysis equipment and disposables industry (Prepared by ANTHONY A. ROMEO). Washington, DC: U.S. GPO, 1984a.
- _____. Medical technology and costs of the medicare program. Washington, DC: U.S. GPO. 1984b.

Medicare's prospective payment system: Strategies for evaluating cost, quality, and medical technology. Washington, DC: U.S. GPO, 1985.

WARR, PETER G. "Pareto Optimal Redistribution and Private Charity," J. Public Econ., Oct. 1982, 18(1), 131-38.

WEISBROD, BURTON A. "Costs and Benefits of Medical Research: The Case of Poliomyelitis, J. Polit. Econ., May-June 1971, 79(3), pp. 527-44.

"The Private Nonprofit Sector: Facts in Search of Theory." Discussion Paper 501, U. of Wisconsin-Madison, Institute for Research on Poverty, 1978.

- WEISBROD, BURTON A. AND DOMINGUEZ, NESTOR O. "Demand for Collective Goods in Private Nonprofit Markets: Can Fundraising Expenditures Help Overcome Free-Rider Behavior?" J. Public Econ., June 1986, 30(1), pp. 83–95.
- WEISBROD, BURTON A. AND MAUSER, ELIZABETII. "Partial vs. Total Effects of Institutional Form: Disentangling Endogenous from Exogenous Variables." Working Paper, Northwestern U., Center for Urban Affairs and Policy Research, 1990.
- WEISBROD, BURTON A. AND SCHLESINGER, MARK. "Public, Private, Nonprofit Ownership and the Response to Asymmetric Information: The Case of Nursing Homes," in *The economics of nonprofit institutions*. Ed.: SUSAN ROSE-ACKERMAN. NY: Oxford U. Press, 1986, pp. 133–51.
- WILENSKY, GAIL. "Technology as Culprit and Benefactor." Paper presented at conference on Health Care Policy at the College of William and Mary, Nov. 12–14, 1987.
- WILENSKY, GAIL AND ROSSITER, LOUIS. "A Re-Examination of the Use of Physician Services: The Role of Physician Initiated Demand." *Inquiry*, Summer

WILLIAMSON, OLIVER. The economics of discretionary behavior. Chicago: Markham Press, 1967.

WOLFE, BARBARA. "Assessment of the Effect of Medicare's PPS on the Demand for V.A. Inpatient Services: An Examination of Externalities of a Public Policy." Paper presented at the meetings of the Association for Public Policy and Management (APPAM), Washington, DC, Nov. 1989.

WOODWARD, ROBERT S. AND WARREN-BOULTON.

FREDERICK. "Considering the Effects of Financial Incentives and Professional Ethics on 'Appropriate' Medical Care," J. Health Econ., Dec. 1984, 3(3), pp. 223–37.

 YOUNG, DENNIS. If not for profit, for what? Lexington, MA: Lexington Books, D. C. Heath, 1983.
ZUCKERMAN, STEPHEN. "Medical Malpractice: Claims, Legal Costs, and the Practice of Defensive Medicine," Health Affairs. Fall 1984, 3(3), pp. 128-33. Journal of Economic Literature Vol. XXIX (June 1991), pp. 553–572

A Review Essay on Handbook of Industrial Organization*

By ROBERT H. PORTER Northwestern University

I am grateful to Kyle Bagwell, Kiminore Matsuyama, Steven Matthews, Therese McGuire, John Panzar, Martin Perry, Julio Rotemberg, F. M. Scherer, Steven Wiggins, and an anonymous referee for helpful comments, and to the National Science Foundation for financial support, via grant No. SES-8721231.

Introduction

THIS ARTICLE critically reviews the Handbook of Industrial Organization (henceforward the "Handbook"), edited by Richard Schmalensee and Robert Willig. These two volumes are the tenth installment in the North-Holland Handbooks in Economics series, under the general editorship of Kenneth Arrow and Michael Intriligator. Like its predecessors, this Handbook contains a number of survey papers (in this instance, 26) on a variety of related topics. As such, they afford both authors and readers an opportunity to determine which directions research in the field has taken, what (if any) real advances have been made, and what questions are still unanswered. Consequently, this review also describes and appraises the current state of Industrial Organization.

Research in Industrial Organization has undergone a dramatic change in the last 20 years. Neoclassical decision-theoretic analysis and competitive general equilibrium theory have been supplanted almost completely by noncooperative game theory. This change was not merely the adoption of the tools of another field.

* Handbook of Industrial Organization. Edited by Richard Schmalensee and Robert D. Willig. 2 vols. Amsterdam and NY: North-Holland, 1989. Pp. 1,620, approx. \$157. ISBN 0-444-70436-1, set. Theoretical modeling issues in Industrial Organization have influenced research in economic theory and game theory, often capturing the attention of leading scholars in these fields. Most of the authors in the Handbook discuss these theoretical developments only when they are pertinent to debates in the field, and tend to shy away from discussions of more technical results. Similarly, I shall concentrate on theories that have affected our understanding of industrial markets.

Research in Industrial Organization used to be categorized into two camps, centered in Harvard and Chicago. One aspect of this division, apart from disagreement about the efficacy of unregulated market outcomes, was the methodological insistence of the Chicago camp on equilibrium analysis, which often meant competitive or monopoly equilibrium analysis. For example, in 1968 George Stigler made the famous (or notorious) claim that "there is no such subject as Industrial Organization," and that the questions addressed are "precisely the content of . . . price or resource allocation theory." Stigler also described the literature as "nontheoretical, or even antitheoretical" (1968, p. 1). The nontheoretical views of the time may not have been well formulated, but they arose because behavior in many industrial markets was seen as inconsistent with existing equilib-